Dear Sir or Madam,

thank you for your helpful comments on the format of our manuscript. We checked the manuscript for any so far undetected typographical errors and corrected them. Further on we accounted for your comments as following:

Major revisions

"Authors Contributions: Please state that all authors read and approved the final manuscript."

We added this very correct statement.

"Table: The table at the end of the manuscript is labelled 'Table 2'. Please correct this."

This is surprising for us: in our version it is labelled "table 1". Please let us know if we have overlooked something.

Minor revisions

"Footnotes: Please remove these from the title page."

This is again surprising for us as there are no footnotes on the title page. It was produced followin the Word-template on http://www.biomedcentral.com/bmcpsychiatry/ifora/#template

"Introduction: Please rename this section 'Background'."

The section has been renamed.

If there are any questions left please contact me,
Best wishes,
Michael Obermeier